Massachusetts Institute of Technology Handout 26 6.042J/18.062J: Mathematics for Computer Science November 5, 1996 Professors Nancy Lynch and Tom Leighton

Quiz 2 Information

When: Wednesday, November 13, 7:00 to 9:00 PM

Where: 54-100

Crib Sheet: The quiz is not open-book, but you are allowed to bring two

2-sided sheets to the exam. You may have whatever information you like on it but it must be handwritten and must not be xeroxed.

Syllabus: Everything done until and including the recitation on November 8.

Special office hours: There will be special office hours during quiz-week as

follows: Mojdeh : T uesday, November 12, 9 � 11 NE43 � 369 Alex : F riday, November 8, 4 � 6 NE43 � 308 Dimitri : T uesday, November 12, 6 � 8 10 � 178 Gunnar : W ednesday, November 13, 4 � 6 NE43 � 370 Y ong Y eow : W ednesday, November 13, 9 � 11 NE43 � 369